Author's response to reviews

Title: The expression of microRNA-375 in plasma and tissue is matched in human colorectal cancer

Authors:

Lingling Xu (xulingling05@sina.com)
Minzhe Li (leeminzhe@hotmail.com)
Min Wang (damy1988@163.com)
Dong Yan (yd15yt88@163.com)
Guosheng Feng (fgs010bjcyh@126.com)
Guangyu An (anguangyu@hotmail.com)

Version: 2 Date: 13 May 2014

Author's response to reviews:

In the revised version, line numbering is included in the main text. Other parts of the manuscript have not changed.